
\documentclass[10pt]{amsart}
\usepackage[utf8]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{hyperref}

\def\udcs{512.542} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{20F16} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]  %Headings in Russian
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 16, стр. 144--144 (2019) \hspace{65mm}{\rm\small УДК \udcs}} %Volume, pages in Russian,
                                                                                    %filled by Editorial board
     \newline
         {\rm\small DOI~10.33048/semi.2019.16.xxx}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{1}


    \bibitem{tho1968}
	J.G. Thompson,
	{\it Nonsolvable finite groups all of whose local subgroups are solvable},
	Bull. AMS, \textbf{74} (1968), 383--437.
	
	\bibitem{atlas}
	J.H. Conway, R.T. Curtis, S.P. Norton, R.A. Parker,  R.A. Wilson,
	{\it An atlas of finite groups},
	 Clarendon Press, Oxford, 1985.
	
	\bibitem{hup}
	B.Huppert,
	{\it Endliche Gruppen. I.},
	Springer Verlag, Berlin, Heidelberg, New York, 1979.
	
	\bibitem{maz1993e}
	V.D. Mazurov,
{\it Minimal permutation representation of finite simple classical groups.
	  Special linear, symplectic, and unitary groups},
	Algebra and Logic, \textbf{32}:3 (1993), 142--153.
	
	\bibitem{brh}
	J.N. Bray, D.F. Holt, C.M. Roney-Dougal,
	{\it The Maximal Subgroups of the Low-Dimensional Finite Classical
	  Groups}, 
Cambridge University Press, Lond. Math. Soc. Lect. Note Ser., 2013.
	
	\bibitem{vama1994e}
	A.V. Vasil'ev, V.D. Mazurov,
	{\it Minimal permutation representations of finite simple orthogonal
	  groups},
	  Algebra and Logic,
	 \textbf{33}:6 (1994), 337--350.
	
	\bibitem{va1996e}
	A.V. Vasil'ev,
{\it Minimal permutation representations of finite simple exceptional
	  groups of types $G_2$ and $F_4$},
	  Algebra and Logic, \textbf{35}:6 (1996), 371--383.
	
	\bibitem{va1997e}
	A.V. Vasil'ev,
	{\it Minimal permutation representations of finite simple exceptional
	  groups of types $E_6$, $E_7$ and $E_8$},
	  Algebra and Logic, \textbf{36}:5 (1997), 302--310.
	
	\bibitem{va1998e}
	A.V. Vasil'ev,
	{\it Minimal permutation representations of finite simple exceptional
	  twisted groups},
	  Algebra and Logic, \textbf{37}:1 (1998), 9--20.

    % \bibitem{arx} D.V. Lytkina, A.Kh. Zhurtov, Finite groups whose maximal subgroups have only soluble proper subgroups, \href{https://arxiv.org/abs/2112.09317}{arXiv:2112.09317
    % }, 2021.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}


\thispagestyle{empty}

\title[Finite groups with soluble 2-maximal subgroups]{Finite groups whose maximal subgroups have only soluble proper subgroups}
\author{{D.V. Lytkina, A.Kh. Zhurtov}}%
\address{Daria Viktorovna Lytkina
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} pr. Koptyuga, 4,
\newline\hphantom{iii} 630090, Novosibirsk, Russia
\newline\hphantom{iii} Novosibirsk State University,
\newline\hphantom{iii} Pirogova str., 2,
\newline\hphantom{iii} 630090, Novosibirsk, Russia
\newline\hphantom{iii} Siberian state university of telecommunications and information sciences,
\newline\hphantom{iii} Kirova str., 86,
\newline\hphantom{iii} 630102, Novosibirsk, Russia}%
\email{daria.lytkin@gmail.com}%
\address{Archil Khazeshovich Zhurtov
\newline\hphantom{iii} Kabardin-Balkarian State University,
\newline\hphantom{iii} Chernyshevskogo str., 173,
\newline\hphantom{iii} 360004, Nalchik, Russia}%
\email{zhurtov\_a@mail.ru}%

\thanks{\sc Lytkina D.V., Zhurtov A.Kh.,
Finite groups whose maximal subgroups have only soluble proper subgroups}
\thanks{\copyright \ 2022 Lytkina D.V., Zhurtov A.Kh.}
\thanks{\rm The work of the first author is supported by RFBR, project No. 20-51-00007.}
\thanks{\it Received  January, 1, 2015, published  March, 1,  2015.}%


\semrtop \vspace{1cm}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } We give a description of a finite group whose maximal subgroups possess only soluble proper subgroups, which implies the answer to the well-known question on composition factors of finite groups, whose second maximal subgroups are soluble.\medskip

\noindent{\bf Keywords:} finite group, maximal subgroup, solubility.
 \end{quote}
}


\section{Introduction}

In 1968 J.~Thompson published a paper \cite{tho1968} containing classification of minimal simple groups by proving the fact that a finite simple nonabelian group, all of whose proper subgroups are soluble, is isomorphic to one of the groups of the following list:

{\bf List 1.} \emph{Minimal simple nonabelian groups}:
\begin{enumerate}
  \item $L_2(2^p)$, $p$ is a prime;
  \item $L_2(3^p)$, $p$ is an odd prime;
  \item $L_2(p)$, $p$ is a prime, $p>3$ and $p\equiv\pm 2\pmod{5}$;
  \item $Sz(2^p)$, $p$ is an odd prime;
  \item $L_3(3)$.
\end{enumerate}

NOTE. Throughout, we use the notation from \cite{atlas}.

The goal of the paper is to get a description of a finite group whose maximal subgroups possess only soluble proper subgroups.

Of course, the desired result will generalize the mentioned classification theorem by Thompson. The only difference from the original study is that we use the classification theorem on finite simple groups stating that every nonabelian finite simple group is isomorphic to one of the groups from the following

{\bf List 2.} \emph{Nonabelian finite simple groups}:
\begin{enumerate}
  \item One of 27 sporadic groups, including the Tits group $^2F_4(2)'$;
  \item $A_n$, $n\geq 5$;
  \item $L_n(q)$, $n\geq 2$, $(n,q)\notin\{(2,2),(2,3),(2,4),(2,5),(2,9),(3,2),(4,2)\}$;
  \item $U_n(q)$, $n\geq 3$, $(n,q)\neq(3,2)$;
  \item $S_n(q)$, $n\geq 4$, $n$ is even, $(n,q)\notin\{(4,2),(4,3)\}$;
  \item $O_{2n-1}(q)$, $O_{2n}^+(q)$, $O_{2n}^-(q)$, $n\geq 4$;
  \item $G_2(q)$, $q>2$;
  \item $F_4(q)$, $E_6(q)$, $E_7(q)$, $E_8(q)$;
  \item $Sz(2^{2n+1})$; $^2G_2(3^{2n+1})$; $^2F_4(2^{2n+1})$, $n\geq 1$; $^3D_4(q)$; $^2E_6(q)$.
\end{enumerate}

Note that the list is designed to avoid repetitions. For example, part 3 does not include groups $L_2(4)$ and $L_2(5)$ isomorphic with $A_5$, $L_2(9)$ isomorphic with $A_6$, $L_4(2)$ isomorphic with $A_8$ and $L_3(2)$ isomorphic with $L_2(7)$.

\section{Main result}

\begin{theorem}
Suppose that $G$ is a finite group. Then all proper subgroups of any of its maximal subgroups are soluble if and only if one of the following conditions hold:

$(1)$ $G$ is soluble;

$(2)$ $G/\Phi(G)$ is a minimal simple group, i.\,e. it is isomorphic to one of the groups from List $1$.

$(3)$ $G$ contains a normal subgroup $G_0$ of prime index, and $G_0/\Phi(G_0)$ is a minimal simple group.

$(4)$ $G/\Phi(G)$ is a simple group, isomorphic to one of the groups from the following List $3$.
\end{theorem}

{\bf List 3.} 
\begin{enumerate}
	\item $L_2(2^{rs})$, $r$ and $s$ are primes;
	\item $L_2(3^{rs})$, $r$ and $s$ are odd primes;
	\item $L_2(p)$, $p$ is a prime and $p\equiv\pm1\pmod 5$;
	\item $L_2(p^r)$, $r$ is an odd prime, $p=5$ or $p$ is a prime, such that $p\equiv\pm2\pmod5$;
	\item $L_2(9)\simeq A_6$;
	\item $U_3(3)$; 
	\item $Sz(2^{rs})$, $r$ and $s$ are odd primes.
\end{enumerate}

\begin{proof} Suppose that case (1) is not true, i.\,e. $G$ is insoluble. Suppose first that every of its proper subgroups is soluble. Let $N$ be a maximal normal subgroup of $G$. Then $\overline{G}=G/N$ is a simple group and $N$ is soluble. Therefore, $\overline{G}$ is nonabelian, all of its proper subgroups are soluble, and $\overline{G}$ is isomorphic to one of the groups from List 1. If $M$ is a maximal subgroup of $G$, then $MN\neq G$ and hence $N\le M$. So, $N$ is contained in the intersection $\Phi(G)$ of maximal subgroups and case (2) is true.

Next suppose that $G$ has an insoluble maximal subgroup $M$. If $M\trianglelefteq G$, then $G/M$ is a group of prime order, all proper subgroups of $M$ are soluble, and case (3) is true.

And if $G$ has no proper normal insoluble subgroup and $N$ is a maximal in $G$ normal soluble subgroup, then $E=G/N$ is a simple nonabelian group and among its maximal subgroups there is an insoluble subgroup $M$. By condition all proper subgroups in $M$ are soluble and, therefore, $M/\Phi(M)$ is a simple group isomorphic to one of the groups from List 1.

If $MN=G$, then $G/N\simeq M/N\cap M$. Since all proper subgroups of $M$ are soluble and $G$ has exactly one insoluble composition factor, then all proper subgroups of $G$ are soluble and, using what we proved before, $G/\Phi(G)$ is a simple group from List 1. So, $N$ lies in any maximal subgroup of the group $G$, i.e. $N\le\Phi(G)$.

Now, to prove the theorem, it is suffice to confirm the following statement.

{\bf Proposition.} {\it Let $G$ be a finite simple group containing a proper insoluble subgroup, and let every such insoluble subgroup modulo its Frattini subgroup is isomorphic to a group from List $1$. Then $G$ is isomorphic to one of the groups from List $3$.}

Since $G$ is isomorphic to one of the groups from List 2, we have to inspect this list.

Information about maximal subgroups of sporadic groups provided in \cite{atlas} shows that none of these groups fits to be the group $G$.

Suppose that $G=L_n(q)$, $q=p^s$, $p$ is a prime, $s\in\mathbb{N}$. If $n=2$, then by \cite[Satz II.8.27]{hup} $G$ is isomorphic to one of the following groups:
\begin{itemize}
	\item $L_2(2^{rs})$, $r$, $s$ are primes;
	\item $L_2(3^{rs})$, $r$, $s$ are odd primes;
	\item $L_2(p)$, $p\equiv\pm1 \pmod 5$;
	\item $L_2(p^r)$, $p=5$ or $p\equiv\pm2 \pmod 5$, $r$ is an odd prime;
	\item $L_2(9)\simeq A_6$.
\end{itemize}
Note, along the way, that none of the alternating groups $A_m$ for $m\ge 7$ satisfies the conditions of the theorem.

If $n\ge3$, then the stabilizer $S$ in $G$ of a one-dimensional subspace of projective space $P_{n-1}$ under natural action of $G$ is isomorphic to $\left([q^{n-1}]:GL_{n-1}(q)\right)/(q-1,n)$. If $S$ is insoluble, then it does not satisfy the conditions of the proposition. It is soluble given $n=3$ and $q=2$ or $3$. For these two cases $G$ is in List 1.

Another case is when $G=S_n(q)$, $n\ge 4$, $n$ is even, $(n,q)\notin\{(4,2),(4,3)\}$. Then the maximal subgroup of the smallest index in $G$ is insoluble by \cite{maz1993e} and does not satisfy the conditions of the Proposition.

Let $G=U_n(q)$. 

If $n=3$, then Tables 8.5 and 8.6 from \cite{brh} show that the conditions of the proposition are met only by $G=U_3(3)$.

If $n=4$, then one of the maximal 2-local subgroups of the group $U_4(q)$, which is isomorphic to $\left([q^4]:SL_2(q^2):(q-1)\right)/(q+1,4)$ is insoluble and does not satisfy the conditions of the proposition.

Now, if $n\ge 5$, then the monomial subgroup of $U_n(q)$ isomorphic to\linebreak $\left((q+1)^{n-1}/(q+1,n)\right):S_n$ is insoluble and does not meet the conditions of the proposition.

If $G$ is an orthogonal group in dimension at least 7, its subgroup of the smallest index is insoluble and does not satisfy the conditions of the proposition (see \cite{vama1994e}).

The same is true for exceptional groups of types $F_4$, $E_6$, $E_7$, and $E_8$, and also for groups of types $^2E_6$ and $^2F_4$ (see \cite{va1996e,va1997e,va1998e}). Maximal subgroups of simple groups of types $G_2$, $Sz$, $^2G_2$, and $^3D_4$ are listed in \cite{brh} (Tables 8.16, 8.30, 8.41, 8.42, 8.43, 8.51). Except for the groups $Sz(2^{2n+1})$, none of the groups in those tables satisfies the conditions of the proposition.

For Suzuki groups, the conditions of the proposition are met only by the groups $Sz(2^{rs})$, where $r$ and $s$ are odd primes. This proves the proposition along with the theorem.
\end{proof}

The theorem ovbiously implies the answer to the known question on composition factors of finite groups, whose second maximal subgroups are soluble.

{\bf Corollary.} {\it Consider a finite group $G$, such that every maximal subgroup of its maximal subgroup is soluble. If $C$ is a composition factor of $G$, then $C$ is either a group of prime order, or $C$ is isomorphic to one of the groups from List $1$ or List~$3$.}

The earlier version of the paper was published on arXiv e-print repository (arXiv:2112.09317).

\bigskip

\end{document}
